Author's response to reviews

Title: Avoidance behaviors and negative psychological responses in the general population in the initial stage of the H1N1 pandemic in Hong Kong

Authors:

Joseph TF Lau (jlau@cuhk.edu.hk)
Sian M Griffiths (siangriffiths@cuhk.edu.hk)
Kai-chow Choi (kchoi@cuhk.edu.hk)
Hi Yi Tsui (hytsui@cuhk.edu.hk)

Version: 6 Date: 14 May 2010

Author's response to reviews:

Dear Dr. Norton,

Thank you once again for your interests in our paper titled “Avoidance behaviors and negative psychological responses in the general population in the initial stage of the H1N1 pandemic in Hong Kong”. We would like to express our gratitude once again to the reviewers taking time from their already busy schedules to provide commentary on the manuscript. We have revised the manuscript with full account of the comments by the reviewers.

Looking forward to hearing from you soon.

Yours sincerely,

Prof. Joseph T. F. Lau
Associate Director, School of Public Health and Primary Care
Director, Centre for Health Behaviours Research
School of Public Health and Primary Care, Faculty of Medicine
The Chinese University of Hong Kong
5/F., School of Public Health and Primary Care
Prince of Wales Hospital, Shatin, NT
Hong Kong
Tel: (852) 2252-8727
Fax: (852) 2645-3098
e-mail: jlaucuhk.edu.hk